Author’s response to reviews

Title: Rationale and protocol for the Assessment of Impact of Real-time Continuous Glucose Monitoring on People Presenting with Severe Hypoglycaemia (AIR-CGM) Study

Authors:

Parizad Avari (p.avari@imperial.ac.uk)

Rozana Ramli (r.ramli@imperial.ac.uk)

Monika Reddy (m.reddy@imperial.ac.uk)

Nick Oliver (nick.oliver@imperial.ac.uk)

Rachael Fothergill (rachael.Fothergill@lond-amb.nhs.uk)

Version: 3 Date: 07 Oct 2019

Author’s response to reviews:

We thank the editor and the reviewers for their comments and suggestions to improve our manuscript. Please find below our responses to the comments highlighted in bold.

Editor comments:

1. In your 'Ethics approval and consent to participate' section, please explicitly state if the ethics committee approved your study.

   – Updated. The study has been given ethical approval for conduct in the NHS by the Health Research Authority - London Bromley Research Ethics Committee (reference 18/LO/1525).

2. The specific individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:

   http://www.biomedcentral.com/submissions/editorial-policies#authorship

Currently, the contributions of author Rozana Ramli (RR) are missing.

   - Updated. RR contributed towards the first draft of the manuscript and has read and approved the final manuscript.
3. Please remove the SPIRIT document from the end of your manuscript file as this is no longer needed in the publication process

   – Done

4. Please confirm whether the image provided in your supplementary figure is your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it. If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time. Should you require an alternative source you may wish to try Wikimedia Common.

   – Thanks, we have excluded this figure as it is described in the text, and is readily available online.

5. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

   - Done, thank you.